Author's response to reviews

Title: Tachykinin receptors antagonism for asthma: a systematic review

Authors:

Renata Ramalho (rrenata@med.up.pt)
Raquel Soares (ragsoa@med.up.pt)
Nuno Couto (nuno.couto@gmail.com)
André Moreira (andremoreira@med.up.pt)

Version: 2 Date: 29 September 2010

Author's response to reviews: see over
29th September 2010

Dear Dr. Melissa Norton:

Please find enclosed the changed manuscript of our original article (manuscript MS: 8281822704403875) entitled “Tachykinin receptors antagonism for asthma: a systematic review” to be considered for publication as an article in the BMC Pulmonary Medicine. All authors will take the public responsibility for the content of the manuscript and provide any relevant data upon request.

On line 134 we clarified that nineteen studies seemed to be suitable based on their title. Based on the abstract, twelve studies failed to meet the inclusion criteria. The full text of seven studies was obtained and they were included in our review. We adhered to the PRISMA guidelines and we added a statement to that effect within line 142. The author’s contributions were added in line 241.

This article represents original work by the authors. No portion of the work has been or is currently under consideration for publication elsewhere and no portion of the work has been published or posted on the internet. All authors have read and approved this final manuscript.

We hope you will consider the manuscript suitable for publication in the BMC Pulmonary Medicine. We are also prepared to further improve the manuscript according to your suggestions.

Yours sincerely,

Renata Ramalho, MSc
Department of Immunology
Faculty of Medicine, University of Porto
Porto, Portugal